Author’s response to reviews

Title: Predictors of intra-hospital mortality in patients with diabetic foot ulcers in Nigeria: data from the MEDFUN Study.

Authors:

OLUFUNMILAYO OLUBUSOLA ADELEYE (funmiadeleye2004@yahoo.com)
Ejiofor T Ugwu (ofornet@yahoo.com)
Ibrahim D Gezawa (gezawa01@yahoo.com)
Innocent Okpe (innokpe@yahoo.com)
Ignatius Ezeani (ignatiusez@yahoo.com)
Marcelina Enamino (marcelina.enamino@yahoo.com)

Version: 5 Date: 20 Aug 2020

Author’s response to reviews:

20/08/2020

The Editor,
BMC Endocrine Disorders

Dear Sir/Madam,

BEND-D-19-00404R4
Olufunmilayo O Adeleye, MBBS, FWACP,FACE; Ejiofor T Ugwu, MBBS FWACP FACE; Ibrahim D Gezawa, MBBS FMCP FRCP; Innocent Okpe; Ignatius Ezeani; Marcelina Enamino
BMC Endocrine Disorders

Thank you for accepting to publish our manuscript in your highly esteemed journal.

The following is a point by point response to the issues raised

1. The full names of the Ethics committees have been included.
2. The textual overlap in the two previously published articles has been completely rephrased.
Sincerely,

Dr Olufunmilayo Adeleye
Corresponding Author.